Notes of the RICH1 Phone meeting 31/03/2006
People Present:
Bristol:  Richard
CERN: Carmelo (part), Christophe, Fabio, Nick, Olav(part), Werner(part) 
IC:        Bill, Dave, Trevor
Oxford: Neville(part)
RAL:     Derek, Mike
Apologies for absence:
 Oxford: Rohan, Tony    
 Milano: Clara   
Discussed Agenda Topics:
1. Gas Enclosure 
1.1 Mounts: The mounts for the installation of the GE in the magnetic shield enclosure will be in CERN next Monday. A test installation on the trolley will be done when the crane is available.

1.2 Installation Hoist and other parts:  Rohan’s mail:  “I have started working on the HPD mounting/removal system and circulated a conceptual design. Hope to meet the Imperial team next week. All the other LHCb design work is complete. Tony is getting the installation tools together.” 
1.3 Painting:   Milano tested the effect of the RICH2 paint samples on aerogel (~2 weeks). The clarity of the aerogel was the same but the refractive index changed slightly (~2( effect). They will conduct aerogel compatibility tests with the polyurethane paint samples next week and continue with the RICH2 exposure to monitor the refractive index.
1.4 Survey
The GE should be surveyed after Easter. The surveyors will be contacted (action Fabio).

1.5 Platforms
The platform on the cryogenic side has no fixed installation date yet. It is not needed for the GE installation but an estimate of its installation date needs would aid planning the installation of other items (action Werner).

2. VELO Seal
It is packed up in its crate and it is ready to move. Design work for the installation kit has begun, everything fits fine.  End of next week will ask for quotations.

3. Exit Window
The central part of the EW is under investigation: how to manufacture the lip of the diaphragm. 

Diaphragm:  Some other darker samples (double pigment and pure black silicone) have been produced and sent to IC for testing. The pure black pigmented silicone has the worst mechanical properties.

Need to address the installation of the EW with Delio (action Trevor).
4. Spherical Mirrors
Two constructive phone meetings with CMA took place in the last two weeks. The mirrors are designed to be made up of two carbon fiber skins separated by 25 mm deep hollow cylinders 50 mm in diameter. The mirror support is also carbon fiber. The mirrors are supported and adjusted using 3 adjusters mounted at the 2 outer edges. Two C-frames fixed to the GE by a rail system support the whole mirror system. The rail system is an IC responsibility. Close working interaction with CMA is needed. Geoff will visit CMA in ~4 weeks to discuss the mirror supports. Things with CMA are progressing positively and in the right direction.   It has been agreed with CMA to have a regular weekly phone meeting, Wednesdays at 17:30.
The demonstration mirror will be built once the drawings showing the mounting points for it are completed. The CMA samples should be in CERN within 1-2 weeks. Plan to conduct fluorocarbon and radiation testing on them (action Fabio).  A sample for the CERN fire testing will also be delivered.
5. Flat mirrors
Milan brought 19 flat mirrors to CERN yesterday. Two mirrors are slightly damaged at the side. In two weeks Milan will come bring another 5 mirrors. The mirrors will be measured (action Fabio and Carmelo). The reflective coating will be during April
Retainer Ring:  The pieces to make the tests are ready. If the tests are successful the order for the final pieces will be done towards the end of next week (~3 weeks delivery)
Mechanics:  The parts have been machined at the IC workshop. The repeatability of disassembly/reassembly was not adequate and the mechanics have been modified but the measurements on the CMM are not yet done (next week).  

6. HPD Assembly
Last week there was a meeting with Rohan to discuss the rail system for access and maintenance of the HPDs.

7.  Quartz Windows
Coating planned for end of this year. The order is not yet out (~6 months delivery time). The final dimensions still need to be defined, but should be ready soon.
8.  LAMS

 A radiation tolerant CID camera has been ordered for testing. It should arrive in a week. 

9.  Magnetic Distortions

The Syracuse system which allows an absolute positioning measurement requires an optical fiber x-y transport stage in front of the HPDs. The allocated volume in the photon funnel for the Syracuse system is difficult to accommodate. Regular four-weekly meetings are planned with Syracuse to converge on a solution to the difficulties. Alternative solutions may need to be explored.

10.  Aerogel

Clara’s mail:

“1. We measured the aerogel left since 2 weeks in the sealed box painted with the RICH2 paint. The result is: CLARITY has not changed at all, index of refraction has increased by 0.0008 (about twice the error with which n is measured). We have put back the tile in the box, and we will monitor if n keeps going up or if it stabilizes.

2. We put in another box an aerogel tile with the samples of paints Chris gave to me Wednesday.

3. We will go to Parma on the 11th April, so we will have in hands the final aerogel containers by then.

3. Roberto made a first drawing for a carbon fiber screen for the VELO, screwed under the aerogel rails. We will send it to you to have a look. If you find it is a good solution, we could ask the firm in Parma to realize it.”

11. Next meeting
Thursday 13th of April 2006 at 10:00(GMT) 11:00(CET). 
